

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Boubakri, Narjess; Cosset, Jean-Claude; Guedhami, Omrane

Working Paper

Postprivatization Corporate Governance: the Role of Ownership Structure and Investor Protection

Nota di Lavoro, No. 37.2003

Provided in Cooperation with:

Fondazione Eni Enrico Mattei (FEEM)

Suggested Citation: Boubakri, Narjess; Cosset, Jean-Claude; Guedhami, Omrane (2003): Postprivatization Corporate Governance: the Role of Ownership Structure and Investor Protection, Nota di Lavoro, No. 37.2003, Fondazione Eni Enrico Mattei (FEEM), Milano

This Version is available at: https://hdl.handle.net/10419/118054

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Postprivatization Corporate Governance: the Role of Ownership Structure and Investor Protection

Narjess Boubakri, Jean-Claude Cosset and Omrane Guedhami NOTA DI LAVORO 37.2003

APRIL 2003

PRIV - Privatisation, Regulation, Antitrust

Narjess Boubakri, *Hec Montreal, Montreal, Canada* Jean-Claude Cosset, *Université Laval, Quebec, Canada* Omrane Guedhami, *Université Laval, Quebec, Canada*

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/web/activ/wp.html

Social Science Research Network Electronic Paper Collection: http://papers.ssrn.com/abstract_id=XXXXXX

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei

Postprivatization Corporate Governance: the Role of Ownership Structure and Investor Protection

Summary

We investigate the role of ownership structure and investor protection in postprivatization corporate governance. We find that the government relinquishes control over time, mainly to the benefit of local institutions and foreign investors. We also show that private ownership tends to concentrate over time. In addition to firm-level variables, investor protection, political and social stability explain the cross-firm differences in ownership concentration. We find that the positive effect of ownership concentration on firm performance matters more in countries with weak investor protection and that private domestic ownership leads to higher performance.

Keywords: Corporate governance, privatization, performance

JEL: G32, G38

The authors appreciate helpful comments and suggestions from Martin Boyer, Sylvain Dessy, Klaus Fischer, Nandini Gupta, Anton Miglo, Oumar Sy, and Désiré Vencatachellum on previous versions of this paper. The authors also received valuable comments from seminar participants at Universite Laval, Memorial University, University of Ottawa, University of Regina, the 2002 APFA/PACAP/Financial Management Association Conference in Tokyo, the 2002 Northern Finance Association Meeting in Banff, the Conference on Privatization, Corporate Governance, and Stock Market Development in Milan, and the 2002 Financial Management Association Meeting in San Antonio. The authors are indebted to Martin Desrochers, Souheil Hassine, and Meissa N'Dir for excellent research assistance. All errors remain the authors' responsibility. The authors gratefully acknowledge the financial support of the Social Sciences and Humanities Research Council of Canada.

Address for correspondence:

Narjess Boubakri HEC Montréal 3000, chemin de la Côte-Sainte-Catherine Montréal (Québec) Canada H3T 2A7 Phone: +1-514-340-5648

Fax: 1-514-340-5632

E-mail: Narjess.Boubakri@hec.ca

Postprivatization corporate governance: the role of ownership structure and investor protection

1. Introduction

Corporate governance is defined as a response to the agency problems that arise from the separation of ownership and control in a corporation. In this paper, we examine corporate governance within the context of privatization. Privatization provides an interesting setting in which to understand corporate governance, because it is a discrete event that often leads to a drastic change in the ownership structure. Thus, privatization is a natural experiment to examine how corporate governance mechanisms evolve, interact, and affect firm performance (Denis and McConnell, 2003).

In this study, we investigate the relation between ownership structure, investor protection and firm performance. Specifically, we seek to answer the following questions:

(1) What is the ownership structure that results from privatization and how does it evolve thereafter?

(2) Does the level of investor protection influence the postprivatization ownership structure?

(3) Does the postprivatization ownership structure depend on other factors?

(4) How do ownership structure and investor protection relate to firm performance and what explains this relation?

To our knowledge, our work is the first multinational study that tracks the postprivatization ownership structure and its determinants. Further, we examine how the ownership structure, a key mechanism of corporate governance, affects the postprivatization firm performance.

¹ See Megginson and Netter (2001) and Djankov and Murrell (2002) for comprehensive surveys of empirical studies on privatization.

We consider a multinational sample of 170 newly privatized firms headquartered in 26 emerging markets. For several reasons, emerging markets provide an excellent laboratory for studying the role of ownership and investor protection in postprivatization corporate governance. First, national legal systems vary markedly across emerging markets. For example, most of these markets suffer from a poor legal environment and a weak enforcement of laws (La Porta et al., 1998). Second, markets for corporate control are not well developed in emerging markets (Shleifer and Vishny, 1997). The deficiencies of these external governance mechanisms (i.e., the market for corporate control and the legal system) offer an interesting opportunity to investigate whether the ownership structure, an internal governance mechanism, plays an important role and interacts with the legal system to address the manager/shareholder postprivatization agency problem. Third, governments in most emerging market countries undertook large-scale privatization programs in the 1990s. With the recent trend towards globalization and the growing interest of investors in emerging markets, documenting and understanding postprivatization corporate governance in these countries takes on particular importance.

We describe the ownership structure that emerges from privatization and examine how it evolves along two dimensions, the degree of ownership concentration and the identity of owners. Our findings suggest that following privatization, private ownership tends to concentrate over time. We also show that on average, privatization results in control relinquishment by the government. The decrease in the government ownership is mostly absorbed by local institutions, foreign investors and domestic individuals.

We then investigate the determinants of the postprivatization private ownership concentration. Building upon Demsetz and Lehn (1985) and La Porta et al. (1998), we include firm- and country-level explanatory variables and take into account the potential

selection effects resulting from the privatization process. Based on panel data, our results show that both sets of explanatory variables are important. For example, we find that firm size, sales growth and industry affiliation as well as investor protection, political and social stability explain the cross-firm differences in ownership concentration.

We empirically examine Shleifer and Vishny's (1997) theoretical claim that ownership concentration is an efficient governance mechanism in countries where investor protection is weak, and should therefore lead to an enhancement in firm performance. We find that ownership concentration is significantly and positively related to the postprivatization firm performance. This effect is more (less) pronounced when the level of investor protection is low (high). Our investigation of the association between the identity of owners and the firm performance suggests that performance is higher when local institutions and domestic investors are involved.

Our paper complements the literature that examines the link between the design of privatization programs, postprivatization ownership structure and investor protection. Megginson et al. (2002) and Bortolotti et al. (2000) show that the choice of privatization methods (i.e., public share issues versus private asset sales) varies across countries and is influenced, among other factors, by the national legal environment. Likewise, Dyck and Zingales (2002) show that privatizations are more likely to occur through private sales and ownership is more concentrated in countries where private benefits of control are larger, i.e., countries with weak legal protection. Our evidence complements their findings by showing that ownership concentration is higher in civil law countries. This finding captures the fact that privatization occurs mainly through direct sales in these countries.

The paper is organized as follows. Section 2 describes the relation between investor protection, privatization, and firm performance. Section 3 presents the data used in the

study. Section 4 documents the postprivatization evolution in ownership structure and investigates its determinants. Section 5 examines the link between the ownership and performance of newly privatized firms. Section 6 summarizes our findings.

2. Investor protection, privatization and firm performance

The standard principal-agent problem arises when the owner of the firm (the principal) is not the same as the manager that controls the firm (the agent). This problem is often put forward to explain the poor performance of state-owned enterprises (SOEs). In this case, the separation problem involves the public (owner or taxpayers) and bureaucrats (politicians). Shleifer and Vishny (1997) describe this relationship as a situation in which bureaucrats retain concentrated control rights without cash flow rights, which are dispersed among the taxpayers of the country. In SOEs, the bureaucrats' main concern is to achieve their political objectives, which do not necessarily converge with the profit maximization objective. Privatization transfers ownership to outside investors, who place greater emphasis on profits and efficiency (Boycko et al., 1996; Shleifer and Vishny, 1997).

Nevertheless, the ultimate success of privatization depends on the effectiveness of postprivatization corporate governance mechanisms. The literature generally distinguishes two types of governance mechanisms: internal and external. Internal mechanisms include, among other things, the ownership structure of the firm (e.g., large blockholders). External mechanisms include the legal system, and the labor and takeover markets. Several studies reviewed in Shleifer and Vishny (1997) and Denis and McConnell (2003) examine the role of ownership structure and investor protection (internal and external mechanisms, respectively) in providing efficient corporate governance. Unlike developed countries, many emerging market countries lack an established institutional framework for efficient

corporate governance. Such deficiencies point to the possibility that internal mechanisms may substitute to external mechanisms in providing efficient governance.

In this paper, we focus on the role of two key dimensions of the ownership structure: the ownership concentration and the identity of owners. First, concentrated private ownership is more likely to ensure the success of privatization in countries with low investor protection. Large shareholders, whose wealth depends heavily on firm performance, have more incentives to monitor the managers and ensure that their resources are not diverted. Further, when privatization leads to a diffused ownership structure, the agency costs associated with managerial control may increase even when the costs of political control fall (Boycko et al., 1996).

Second, the identity of owners is also likely to influence the performance of newly privatized firms (NPFs). For example, foreign investors require high information disclosure standards and, for reputation concerns, maintain a strict control of managers' actions (Dyck, 2001). Institutional investors also exert a close monitoring of management activities to ensure superior returns (Boutchkova and Megginson, 2000).

We also examine the relationship between internal (ownership structure) and external (investor protection) governance and the subsequent performance of NPFs. To do so, we use a multinational sample of firms diversified across countries and legal systems.

3. Data and variables

3.1. The sample of privatized firms

To investigate the relation between investor protection, ownership structure, and firm performance, we use a sample of 170 firms privatized in 26 emerging market countries over

the period 1980-1997.² Our sample does not include firms from the ex-communist countries for two reasons. First, the traditional law system in these countries is based on the Soviet law, which has undergone many changes in the transition period (La Porta et al., 2000). Second, the postprivatization ownership structure in these countries is mainly in the hands of insiders (managers and employees).

Table 1 shows that the 170 firms are located in different geographical regions as categorized by the *World Bank*. For example, 27.65 percent are from Africa and the Middle East, 27.06 percent from East and South Asia and the Pacific, 32.35 percent from Latin America, and 12.94 percent from Europe and Central Asia. The diversification across geographic regions is important, because it comprises countries with different legal and institutional environments.

Table 1 also shows that the sample is diversified across industries, with 33.53 percent in the financial sector, 28.24 percent in the energy sector, and 11.18 percent in utilities. Furthermore, 77 percent of the privatizations occurred in the 1990s, which reflects the recent trend towards large-scale privatizations by emerging markets.

When we classify our sample of NPFs by legal origin, we see that 37.65 percent of the firms come from common-law countries and 62.35 percent from civil-law countries. This diversification across legal origins allows us to examine whether the extent of investor protection helps explain the cross-firm differences in the postprivatization ownership structure.³

² Our sample countries are either "major" or "frontier" emerging markets (Emerging Markets Data Base).

³ Our sample largely represents the population of privatized firms in developing countries. By using the *World Bank* list of privatized firms in developing countries, we find that 30.48 percent of the firms

3.2. Ownership variables

To examine the postprivatization ownership structure and its determinants, we focus on two measures of private ownership concentration that are used in the literature (e.g., Demsetz and Lehn, 1985; La Porta et al, 1998; Demsetz and Villalonga, 2001), the percentage of shares held by the three largest private investors, L3, and an approximation of the Herfindahl index (the sum of squared ownership shares by the three largest private investors), H3. We apply a logistic transformation to L3, using the formula log(L3/(1-L3)) to convert a bounded variable to an unbounded one; and a logarithmic transformation to H3. The resulting variables are LL3 and LH3. Throughout this paper, we focus on the LL3 measure of ownership concentration. However, our findings are robust to the use of LH3 as an alternative measure.

To describe the postprivatization ownership structure and its evolution over time, we identify five groups of investors, the government, domestic institutions, foreign investors, employees, and individuals. We define the date of privatization as the date on which the government sold, for the first time, a certain amount of shares. We collect ownership data - from several sources from year –1 to year 3 around the privatization year. The main sources for privatization ownership data are the offering prospectus and annual reports, but we

are from Africa and the Middle East, 17.08 percent from East and South Asia and the Pacific, 42.35 percent from Latin America, and 10.09 percent from Europe and Central Asia. We find that 20.52 percent of the firms are from the financial sector and 15.97 percent are utilities. We also note that 80 percent of the privatization transactions occurred in the nineties. Finally, when we examine the legal origin in the *World Bank* list the distribution shows that 31.12 percent of the firms come from common law countries and 65.27 percent from civil law countries.

also use additional sources such as Worldscope Disclosure, the Asian, Brazilian and Mexican Company Handbooks, as well as the Guide to Asian Companies.

3.3. Performance variables

We measure firm performance by the return on sales (net income to sales), the return on assets (net income to assets), and the return on equity (net income to equity) (Megginson et al., 1994; Boubakri and Cosset, 1998; D'Souza and Megginson, 1999). We obtain financial information from the firms' financial statements, their web sites, and from databases such as Worldscope Disclosure, Global Vantage, and Moody's International. Since we are examining the relation between the postprivatization firm performance and ownership structure, we compute the performance measures for a period of five years (from one year before to three years after privatization including the year of privatization itself). To check whether privatization leads to performance improvements, we also report the three-year average performance before and after privatization.

4. Ownership structure

4.1. Evolution of ownership structure

Panel A of Table 2 reports descriptive statistics on the concentration of ownership measured by the cumulative number of shares of the three largest private investors and the Herfindahl index. The average number of shares held by the three largest shareholders increases from 10.64 percent before privatization to 38.57 percent in the year of privatization. Shareholdings continue to increase at a rate of 6.4 percent per year to 45.96 percent after three years. In comparison, La Porta et al. (1998), using a sample of 49 developed (24) and developing (25) countries, report an average ownership of the three largest shareholders of 46 percent in the 10 largest private nonfinancial firms.

We also report the evolution of the Herfindahl index of ownership concentration. Our results show higher levels of ownership concentration after privatization. The average index increases from 3.36 percent before privatization to 20.58 percent in the year of privatization. It continues to increase to 23.42 percent by the end of the third year. Overall, the results presented here show that the private ownership of NPFs becomes more concentrated over time.

Panel B of Table 2 reports changes in ownership by type of owner. The results indicate a significant change in the ownership structure of the sample firms following privatization. The average government stake declines substantially after privatization. We find a shift of 55 percent in the average government ownership (from 78.16 percent before privatization to 35.23 percent in the privatization year). The average postprivatization government stake continues to decrease over the following three years to 21.41 percent. This finding is consistent with the predominance of partial, staggered sales (Perotti and Guney, 1993; Perotti, 1995). Although the government is the controlling shareholder (more than 50 percent of shares) in 82 percent of the sample firms before privatization, this percentage drops to 34 percent in the privatization year and to 20 percent after three years. Those firms that remain under government control three years after privatization come primarily (46 percent) from strategic sectors (e.g., utilities, telecommunications, airlines, and banking), and from East and South Asia and the Pacific region (82 percent), where partial privatizations are more common. Gupta (2002), for example, provide a comprehensive study of the Indian partial privatization program.

Local institutions absorb much of the decrease in the state ownership. Their average stake increases from 4.68 percent prior to privatization to 19.17 percent in the year of privatization and continues to increase up to 26.59 percent in the third year. These results

indicate the importance of local institutions as key players in the ownership structure of privatized firms. These findings complement Boutchkova and Megginson's (2000) evidence for privatized firms in developed economies.

Foreign investors' average stake doubles in the year of privatization to 14.43 percent and continues to increase in the following three years to 16.26 percent. In a more detailed examination, unreported here, we find that on average, 60 percent of foreign investors are institutions. This finding could reflect the foreign ownership consolidation wave and the inflow of foreign consortiums following the financial liberalization that preceded or accompanied privatization (e.g., Latin American countries).

To ensure employees' support for the privatization policy, many governments allocate a fraction of the share issues (between 5 and 20 percent) to the firm's employees through Employee Stock Ownership Plans (ESOPs). The average employees' stake increases from 0.87 percent before privatization to 3.61 percent in the privatization year. However, we observe a decline in the average employees' stake over the following three years. This result could be explained by the existence of a lock up period during which employees are not allowed to sell their stake (Megginson et al., 1994).

One objective of a privatization program is to widen ownership among individuals. In our sample of privatized firms, individuals' average ownership increases from 3.17 percent before privatization to 14.50 percent in the year of privatization. At the end of the third year, the average stake held by individuals is 15.78 percent. This result suggests that privatization helps create a certain "equity culture" and ensures a minimum level of popular capitalism. For example, privatization in Malaysia has been driven to a large extent by the government's willingness to increase the ownership share of Bumiputras in key sectors of the economy.

4.2. Investor protection and ownership structure

To examine the evolution of the ownership structure according to the origin of commercial laws, we divide the sample into two groups: firms from common law countries and those from civil law countries. The rationale behind this partition is that common law countries provide a better investor protection that may influence the postprivatization ownership structure (La Porta et al., 1998).

Panel A of Table 3 presents the changes in private ownership concentration by legal origin. Using both measures of ownership concentration, we show that the highest postprivatization concentration levels are found in countries with a poor investor protection (civil law countries). In the privatization year, the average stake held by the three largest shareholders is 49.03 percent for firms from civil law countries and 22.67 percent for firms from common law countries. Three years after privatization, the average stake held by the three largest shareholders is 28.64 percent and 58.42 percent for common law and civil law countries, respectively. These postprivatization differences between the common law and civil law countries are statistically significant at the 1 percent level.

Panel B of Table 3 shows how ownership by the government, local institutions, foreign investors, employees, and individuals change and differ according to the legal origin of a given country. The average government stake is 77 percent prior to privatization for firms in common law countries, and 78.86 percent for firms in civil law countries. In contrast, the average government stake after privatization is statistically lower in civil law countries than in common law countries. The average government stake decreases by 50.77 percent in the privatization year for common law countries to 30.82 percent in the third year. For firms from civil law countries, the average government stake decreases by 57.40 percent in the

privatization year and continues to decrease to 15.65 percent in the third year. These results suggest that governments in civil law countries tend to sell higher stakes faster than do those in common law countries.

The evidence on the changes in local institutions' ownership suggests significantly higher levels of institutional ownership in countries with relatively poor investor protection (civil law countries) compared to countries with a better investor protection (common law countries). For the subsample of firms from common law countries the average stake is 13.56 percent in the third postprivatization year and 35.58 percent for the subsample of firms from civil law countries. This result suggests that institutional investors may play a key role in countries with low investor protection.

Before privatization, the average stake held by foreign investors is 13.64 percent for firms from common law countries and 2.37 percent for firms from civil law countries. The difference is statistically significant at the 1 percent level. In the year of privatization, the average foreign stake increases by 10.56 percent for the first subsample and almost fivefold for the second subsample. Three years after privatization, the average stake is 16.05 percent for the common law countries subsample and 16.40 percent for the civil law countries subsample. The difference is not statistically significant. These results suggest that, ceteris paribus, foreign investors are drawn to privatized firms in emerging markets, regardless of the degree of legal protection in a country.

We also explore how employees' ownership is affected by the legal origin. The preprivatization employees' average stake is higher for the subsample of firms from common law countries compared to the subsample of firms from civil law countries. However, this difference is not statistically significant.

We investigate the patterns of individuals' ownership in both common law and civil law countries. Before privatization, the average stake held by individuals is 4.39 percent for firms from common law countries and 2.44 percent for firms from civil law countries. The results indicate that in common law countries, individuals own significantly higher levels for the privatization year and onwards. Three years after privatization, the average stake is 30.39 percent for the common law subsample and 8.11 percent for the civil law subsample. Although we are unable to document the evolution in the number of shareholders, these results suggest that privatization is more likely to create a popular capitalism in common law countries than in civil law countries. In other words, wherever investors are protected by law, we can expect an "equity culture" to emerge and individuals to have more incentives to invest.

< Table 3 >

Overall, the results of Tables 2 and 3 indicate a high concentration of private ownership following privatization and support Shleifer and Vishny's (1997) and Dyck's (2001) contention that privatization should result in higher levels of concentrated ownership in countries with poor investor protection. The following section investigates the determinants of ownership concentration.

4.3. Determinants of ownership concentration

Although the univariate results suggest that privatization yields higher levels of private ownership concentration in countries with a weak investor protection (civil law countries), the results only document bivariate relations.

Demsetz and Lehn (1985) investigate the determinants of ownership concentration in the US. The authors find (1) a negative association between ownership concentration

and firm size; (2) a positive association between ownership concentration and profit instability; and (3) a lower ownership concentration in regulated firms (utilities and financial firms). Recent papers by Himmelberg et al., (1999), Demsetz and Villalonga (2001), and Palia (2001) extend the Demsetz and Lehn analysis, and show that managerial ownership is endogenously determined by a set of firm-level variables in the contracting environment.

La Porta et al. (1998) use country-level explanatory variables to explain ownership concentration in a wide set of developed and emerging markets. These authors measure ownership concentration in each country by the average ownership stake of the three largest shareholders in the ten largest publicly traded companies. They use several measures of legal protection including the quality of law enforcement, the shareholder rights, the creditor rights, and legal origin dummies, and find a negative association between the extent of legal protection and ownership concentration.

While Demsetz and Lehn (1985) and La Porta et al. (1998) examine the determinants of ownership concentration of publicly traded firms, in this research we focus on the determinants of the postprivatization private ownership structure. We extend their empirical specifications by simultaneously including firm- and country-level explanatory variables, and we formulate our conjectures within the following equation

$$LCONC_{it} = \alpha + \beta FLV_{it} + \vartheta CLV_{it} + \gamma_t + \varepsilon_{it}$$
, (1)

where $LCONC_{it}$ is the private ownership concentration measured by LL3, FLV_{it} represents firm-level variables, CLV_{it} represents country-level variables, γ_t are year fixed effects (i.e., an indicator for each postprivatization year) included to capture unobservable changes at the firm- and country-level, and ε_{it} is a disturbance term.

Table 4 summarizes the firm-level and country-level variables we consider. Following Demsetz and Lehn (1985), we use the following firm-level variables displayed in Panel A: Operating risk, for which we use, as a proxy, the standard deviation of the annual return on equity ratios (RISK) during the three years preceding the privatization year; Growth, for which we use, as a proxy, the average annual sales growth rate (SALESGR) during the three preceding years preceding the privatization year; Firm size, for which we use, as a proxy, the natural logarithm of total sales (SIZE); and Industry affiliation. We include a dummy variable for each industry: IND1 for the finance industry, IND2 for utilities, IND3 for telecommunications, IND4 for oil and gas, petrol/petrochemical, cement and mining, and IND5 for other industries.

In addition to the firm-level variables, we control for the timing of privatization. To do so, we include a dummy variable for privatization timing that takes the value of one if the sample firm is privatized after the median privatization date in the country, and zero otherwise (LATE). This variable captures the privatizing government's preferences on the choice of the to-be-privatized firm and the extent of the stake sold. For example, because of social and political costs and for fear of losing revenues, governments could be reluctant to sell higher stakes and relinquish state control at early exploratory stages of the privatization program. Similarly, in an effort to attract private investors, governments might choose to relinquish control in the early stages. The empirical results will validate either one or the other of these arguments.

Panel B presents the country-level variables. Legal system refers to the legal origin of the country. This is a dummy variable that takes the value of one for common law countries and zero otherwise (LAW).⁴ In civil law countries, governments tend to sell higher stakes using direct sales, which is more likely to yield a concentrated ownership (Dyck, 2001; Bortolotti et al., 2000). Thus, we expect a negative association between ownership concentration and ownership protection.

As a proxy for legal enforcement, we use La Porta (1998) index of the efficiency of the judicial system (JUDICIAL). Strong legal enforcement could substitute for the weakness of rules since "active and well-functioning courts can step in and rescue investors abused by the management" (La Porta et al. 1998, p.1140). Thus, we expect a negative correlation between ownership concentration and legal enforcement.

As a proxy for income inequality, we use the Gini index (GINI). Demsetz (1997) contends that a society's wealth and its distribution are likely to explain ownership concentration. Demsetz claims that wealthy people are more likely to buy large stakes while still enjoying diversification of their wealth. Therefore, we expect income inequality to have a positive impact on ownership concentration.

As a proxy for the stability of the political environment, we use the index of government stability (GS) provided by *International Country Risk Guide* (ICRG). This indicator measures the credibility and ability of the government to carry out declared economic reforms and privatization, which have the merit of reducing private investors' worries about future policy reversals (Perotti, 1995; Perotti and van Oijen, 2001). Therefore,

⁴ The exogeneity of the legal origin dummy makes its use as a measure of investor protection appealing in contrast to other measures of investor protection such as the shareholder rights index that might be endogenous to ownership concentration (La Porta et al., 1998; Dyck, 2001).

we expect a positive association between private ownership concentration and political stability.

As a proxy for the stability of the social environment, we use the index of ethnic tensions provided by ICRG (ET). This indicator measures the degree of tension within a country that is attributable to cultural, racial, nationality, or language divisions. These factors are all likely to adversely affect the business environment, private investors' confidence, and their willingness to invest a large portion of their wealth. Therefore, we expect a positive association between private ownership concentration and the stability of the social environment.

< Table 4>

Some of the variables we describe above try to act as proxies for the extent of selection effects inherent to the privatization process. In the context of privatization, we must deal with the likelihood that a government's choices affect the resulting ownership structure. Given the prevailing economic and institutional environment, the privatizing government may privatize (keep) or sell higher (lower) stakes in better-quality firms, which could result in a more (less) concentrated private ownership in these firms. Moreover, the government might be reluctant to relinquish control in large firms and/or in sectors that it believes are economically and politically strategic. Finally, certain types of owners (e.g., local institutions, foreign investors) may be able to identify better-quality firms from institutionally healthier countries. All these cases suggest that the private ownership concentration that emerges after privatization is likely to be systematically related to unobservable firm- and country-characteristics in addition to observable firm- and country-characteristics.

We can control for this unobservable firm and country heterogeneity in panel data. Since our study is multinational, and given that some of our main country-level variables are invariant within firms and countries, we assume that we can capture firm (country) unobservable heterogeneity by introducing industry- (and region-) fixed effects. We believe that a particular firm (country) exhibits the same characteristics as the whole industry (region). Governments generally privatize firms from particular industries using the same timing and sales methods. These practices are also common to countries from a particular region.

Table 5 reports the regression results for estimating Equation (1). We find that the growth variable SALESGR is positive and significantly (at the 1 percent level) related to ownership concentration. This finding is consistent with the fact that firms with high growth prospects are more attractive. We find a negative and significant (at the 1 percent level) relation between ownership concentration and firm size. That is, an increase in the firm's size leads to an increase in the level of ownership dispersion.

Our results also show significantly higher levels of ownership concentration for financial firms. One possible explanation for this finding is that the financial liberalization that accompanied or preceded privatization created a wave of foreign ownership consolidation, particularly in Mexico and other Latin American countries.

Our results also indicate significantly lower levels of ownership concentration for utilities and telecommunications. This evidence supports the previous discussion on the monitoring role of regulation and the reluctance of the government to fully privatize strategic industries. We note that neither the operating risk of the firm nor the privatization timing variable is significantly related to ownership concentration.

At the country level, the results show several significant relations. Specification (1) contains region-fixed effects. Specifications (2) through (4) contain interaction between regions and country-level variables. We find a negative and significant (at the 5 percent level) association between ownership concentration and the level of investor protection across all the specifications. Firms in civil law countries exhibit a level of ownership concentration that is higher than that for firms in common law countries. These results confirm those in Table 3 (Panel A) and suggest that, after controlling for firm and country characteristics, ownership concentration is significantly higher in civil law countries.

The legal enforcement variable JUDICIAL is not a statistically significant determinant of ownership concentration. One possible explanation for this result is that legal enforcement is irrelevant in emerging markets and does not substitute for the weakness of rules. We also find that the measure of income inequality GINI is not significantly related to ownership concentration. However, specification (2) shows that the impact of income inequality is positive and significant (at the 1 percent level) in Latin America and the Caribbean, the region that exhibits the highest income inequality.⁵

As predicted, we find that the government stability GS and ethnic tensions ET variables are key determinants of private ownership concentration. This result suggests that private investors are more inclined to hold larger stakes of firms in more stable political and social environments.

Specifications (3) and (4) indicate that the interactions between regions and government stability GS, and between regions and ethnic tensions ET, are positive and

⁵ Over the 1980-2000 period, the average index of income inequality (GINI) for Latin America and the Carribean is 57, compared to 41 for Africa and the Middle East, 40 for Europe and Central Asia,

and 38 for East and South Asia and the Pacific.

19

significant (at the 1 percent level) for Latin America and both Latin America and Europe, respectively.

In all regressions, none of the year effects is significant. Overall, the results indicate that firm- and country-level explanatory variables explain at least 49 percent of the variation in ownership concentration. These results confirm the endogeneity of private ownership concentration and the necessity to take it into account when testing the ownership-performance relationship.

The results can also be related to the patterns of government ownership reported in Panel B of Table 3, which show higher levels of postprivatization government ownership for firms in common law countries. Governments in emerging markets seem to privatize higher stakes when there is a poor investor protection, which is more likely to lead to a higher private ownership concentration. This result could also explain why privatization methods tend to vary across countries.

The empirical evidence on the privatization methods appears to be mixed. Megginson et al. (2002) show that privatization through private sales is more likely when the investor protection is better (common law countries), but Bortolotti et al. (2000) report a higher frequency of direct sales in civil law countries. Dyck and Zingales (2002) examine the effects of private benefits of control on the development of markets, particularly on the ownership structure of firms and the choice of privatization methods. The authors show that countries that exhibit higher benefits of control, i.e., countries with weak legal protection, have more concentrated ownership. In these countries, privatization is less likely to occur through public offerings.

Our evidence, drawn exclusively from emerging markets, concurs with this line of research on the link between the design of privatization programs, postprivatization ownership structure, and investor protection: we show that postprivatization ownership concentration is higher in countries where investor protection is weak. This evidence indirectly captures the fact that privatization occurs mainly through direct sales in these countries (e.g., in Africa and Latin America).

< Table 5 >

5. Ownership, investor protection, and performance

5.1. Operating performance of privatized Firms

As proxies for the operating performance of NPFs, we use the return on sales (ROS), the return on assets (ROA), and the return on equity (ROE) ratios. Table 6 reports the mean and median levels of performance from one year before to three years after privatization, and the mean and median changes in performance from the preprivatization (years –3 to -1) to the posprivatization (years 1 to 3) periods.

The performance measures tend to increase following privatization. For example, the mean (median) ROS increases from 9.9 (6.4) percent for year –1 to 12.8 (9.6) percent for year 0. Three years after privatization, the mean (median) ROS is 6.2 (9.2) percent. Furthermore, the changes in the performance measures over the window [-3, -1; +1, +3] are significant and positive. For example, the median changes in ROS and ROE are 3.9 percent and 3.7 percent, respectively, but the median change in ROA is 1.1 percent.6

⁶ To account for contemporaneous economy-wide factors that might affect the performance

indicators, we compute market-adjusted measures using a control sample of firms that are privately

5.2. The Relation between ownership concentration, investor protection, and performance

Ownership concentration, by establishing a strong relation between ownership and control, helps mitigate the extent of agency problems in a firm and should yield a superior performance (Shleifer and Vishny, 1997). A standard approach is to conduct a regression analysis of the firm performance on ownership structure variables. However, this method can be misspecified if some of the unobserved determinants of firm performance (unobserved heterogeneity) also explain the ownership concentration variables, leading to a spurious relation between ownership concentration and firm performance. Several recent studies (e.g., Himmelberg et al., 1999; Palia, 2001) document the endogeneity nature of the ownership structure (predominantly managerial ownership) and stress the need to control for it by using instrumental variables for ownership. Accordingly, we estimate the following system of equations

$$PERF_{it} = \delta + \theta_1 LCONC_{it} + \theta_2 LCONC_{it} * PROTEC + \beta_1 FLV_{1it} + \vartheta_1 CLV_{1it} + \gamma_t + \varepsilon_{1it}, \qquad (2a)$$

$$LCONC_{it} = \alpha + \beta_2 FLV_{2it} + \vartheta_2 CLV_{2it} + \varepsilon_{2it},$$
 (2b)

owned over the analysis period. The results, not reported but available from the authors, show that the mean (median) market-adjusted performance before privatization is negative for all indicators, suggesting that previously state-owned enterprises underperform with respect to other firms in the market. However, the mean (median) market-adjusted performance following divestiture becomes positive and significant at the 1 percent level for all three indicators, which confirms the positive impact of privatization on performance.

where $PERF_{ii}$ stands for the performance of firm i (ROS, ROA, ROE) at time t, $LCONC_{ii}$ is the ownership concentration of firm i at time t as measured by LL3, FLV_{1ii} (firm size SIZE and industry dummies) and CLV_{1ii} (economic freedom FREEDOM and GDP growth GDPG) represent firm- and country-level control variables included in the performance equation. Including the GDP growth in the performance equation allows us to indirectly control for the possible impact of economic reforms on firm performance (Bekaert et al., 2001). Including economic freedom allows capturing the cross-country differences in the institutional environment. γ_i are year dummies introduced to control for year specific effects. FLV_{2it} and CLV_{2it} represent firm- and country-level instrumental variables (\mathbf{Z}_{t}) included in the ownership concentration equation, and ε_{1it} and ε_{2it} are the error terms.

The choice of the instruments is a crucial task since they must be highly correlated with ownership concentration but must not determine firm performance. We use the set of instruments from specification (1) in Table 5 (RISK, SALESGR, SIZE, LATE, LAW, GINI, GS, ET). These instruments must satisfy $E[\epsilon_{1it}, Z_{it}] = 0$. Since firm size explains both ownership concentration and firm performance, we use the (logarithm of) total GNP on the ground that larger economies have larger firms and therefore, lower ownership concentration (La Porta et al., 1998).

The parameter θ_1 in Equation (2a) measures the impact of ownership concentration on the performance of privatized firms. We also introduce an interaction term between ownership concentration and investor protection. If the effect of ownership concentration on performance is more pronounced in countries with poor investor protection, then the coefficient of the interaction term (θ_2) should be negative and significant. We measure investor protection using the *PROTEC* variable, which refers to a country's law and order score from ICRG.

Within a two-stage least squares (2SLS) framework, we estimate equations (2a) and (2b). In the first stage, we estimate the equation of the determinants of ownership concentration, LCONC_{it}, and obtain the fitted (estimated) values (with industry- and region-fixed effects). In the second stage, we use these fitted values of ownership concentration as instruments for LCONC_{it} (Equation 2a) to estimate the impact of ownership concentration on firm performance.

We estimate three specifications of Equation (2a) and report the estimation results in Table 7. The results of the regression that restrains the parameter & to zero (specification (i)) suggest that after we control for firm size and industry, cross-country differences in the institutional environment and economic growth, ownership concentration is significantly related to firm performance. This result supports Shleifer and Vishny's (1997) contention that postprivatization diffused ownership structure (and thus an increase in agency costs of managerial control) may lead to disappointing performance. Further, FREEDOM is positive and significantly related to performance, while GDP growth is positively but less significantly correlated with performance. The positive effect of FREEDOM suggests that firms achieve higher performance improvements when they are privatized in an institutionally healthier environment. The negative and significant coefficient estimate for SIZE suggests that small SOEs are more likely to adapt to a change in the environment.

We also control for industry characteristics and year-specific effects. An F-test for testing the null hypothesis of equal effects through industries shows a strong rejection for all specifications. An examination of industry effects indicates a superior performance for

_

⁷ We use ROS as our measure of firm performance. The results with ROA or ROE, which are available from the authors, are qualitatively similar.

noncompetitive firms (i.e., utilities and telecommunications) and for firms from the financial sector. None of the year effects is significant in the three specifications.

We estimate an alternative specification (ii) of equation (2a) that does not restrain to zero the parameter θ_2 to allow for an interaction between ownership concentration and investor protection. We note that performance and ownership concentration continue to be positively and significantly correlated.

More important, we find that the interaction variable LCONC*PROTEC has also a negative and significant impact on performance. This finding suggests that even though concentration has a positive impact on performance, the effect is stronger wherever investor protection is weaker. The result also supports the belief that in emerging markets, ownership concentration can substitute for investor protection and provide the functions of corporate governance.

We also show that PROTEC has a negative and significant effect on performance, suggesting that firms in countries with low investor protection perform better. This result should be interpreted together with the previous findings on the impact of ownership concentration and its interaction with investor protection on firm performance. To compensate for the lack of an efficient legal system, investors in countries with poor investor protection rely on governance structures that are basically dominated by highly concentrated ownership. SIZE has a significant and negative effect on performance. The country-level control variables indicate that FREEDOM remains positive and significant, but GDPG is insignificant.

Several authors provide evidence of nonlinearities in the ownership-performance relation (Morck et al., 1988; McConnell and Servaes, 1990; Claessens and Djankov, 1999;

Himmelberg et al., 1999). To check for this possibility, we extend specification (ii) to include the instrumented LCONC and its squared variable (LCONC)² in specification (iii). We find that the estimated coefficient of the squared term (LCONC)² is positive and significant at the 1 percent level.

Throughout the three specifications, we check whether the error term is uncorrelated with the instrumental variables \mathbf{Z}_{it} , that is, $E[\epsilon_{1it} \ \mathbf{Z}_{it}] = 0$. The *F*-statistics reported in the last column of Table 7 imply that we cannot reject the null hypothesis of no correlation between the error terms and instruments, suggesting that our system of equations is properly identified.

The use of the instrumental variable approach might also suggest that we are measuring the effect of one or more particular instruments on firm performance rather than the effect of ownership concentration on firm performance (Palia, 2001). We check for this possibility by regressing firm performance on the instruments \mathbf{Z}_{ir} . The results, not reported here but available from the authors, show that only the instrument *RISK* is statistically significant at the 5 percent level. The adjusted R^2 of 2.1 percent is very low compared to 26.3 percent obtained in the regression of ownership concentration on these instruments. When we conduct an *F*-test for the joint insignificance of the instruments, we find a rejection at the 5 percent level for the performance regression (*F*-statistic = 2.58, *p*-value = 0.0129) and 1 percent level for the ownership concentration regression (*F*-statistic = 22.44, *p*-value = <.0001). These results suggest that our instrumental variable approach mainly captures the effect of ownership concentration on firm performance.

< Table 7 >

5.3. The relation between ownership identity and performance

So far, we have examined only the relation between ownership structure and firm performance along the ownership concentration dimension. However, the identity of owners could also be relevant in analyzing the postprivatization ownership-performance relation (Boycko et al., 1996; Dyck, 2001). Different types of owners may have distinct incentives and abilities to control the postprivatization managers. Therefore, it is ultimately the extent of this control that determines the firm's performance. Thus, we assess the impact of the different types of owner, as identified in Section 4 on the performance of NPFs.

To do so, we perform a multivariate regression analysis in which we use the shares held by various stakeholders while controlling for the firm's size, and the country's institutional environment and economic growth. We formulate the following set of equations

$$PERF_{it} = \delta + \sum_{j} \theta_{j} OWNER_{ijt} + \beta_{1} FLV_{it} + \vartheta_{1} CLV_{it} + \gamma_{t} + \varepsilon_{1it}, \qquad (3a)$$

$$OWNER_{ijt} = \alpha_2 + \beta_2 FLV_{it} + \vartheta_2 CLV_{it} + \varepsilon_{2it}, \qquad (3b)$$

where $OWNER_{ijt}$ is the percentage of shares held by the owner of type j (j = 1,2,3,4) of firm i at time t.

To estimate the system of equations (3a and 3b), we instrument each type of owner OWNER using the same set of instruments as in equation (2b). Our idea is that the percentage held by certain types of owners could be endogenously determined by public information released from the privatization process.

Panel B of Table 7 reports the results from the estimation of equations (3a and 3b). First, local institutions' ownership has a significant, positive impact on the firm's performance, suggesting that local institutions are an effective mechanism of the postprivatization corporate governance. We also show that ownership by individuals has a significant, positive influence on performance. Taken together, our results suggest that local outsiders provide beneficial restructuring.

Surprisingly, our results suggest no significant association between firm performance and foreign ownership. At first sight, this result is puzzling, because it contradicts the theoretical contentions of Boycko et al. (1996) and Dyck (2001) that foreign investors are a source of better governance and higher performance. Yet, Frydman et al. (1999) find a similar result in that domestic outsiders do better than foreign investors in newly privatized firms in the Czech Republic. The authors argue that perhaps foreign owners are not initially at ease in an environment that is relatively unknown to them, and that the transfer of knowhow may require more time than their sample period allows.

We find no significant relation between the firm's performance and employee ownership. To some extent, this result does not disagree with Boycko et al's (1996) prediction that employees make poor stockholders/monitors. For the control variables, we find that the levels of institutional and economic development are positively related to firm performance. None of the time effects is significant. The *F*-statistics reported in the last column of Panel B indicate that we cannot reject the null hypothesis of no correlation between the error terms and the instruments.

6. Conclusion

In this paper, we investigate the relation between investor protection, ownership structure, and firm performance. Our sample comprises 170 firms from 26 emerging markets that were privatized during the 1980-97 period.

First, our results show a significant decrease in the government ownership in the year of privatization; the postprivatization government share continues to decrease in the following three years. Much of the decrease in the government ownership is absorbed by local institutions, foreign investors and individuals.

Second, we find that firm-level variables, i.e., size, sales growth, and industry affiliation, and country-level variables, i.e., investor protection, and political and social stability, are key variables in explaining the cross-firm differences in ownership concentration.

Finally, using a methodology that controls for the endogeneity of ownership, we show that firm ownership concentration is positively related to firm performance. We find that the effect of ownership concentration on firm performance is stronger in those countries where investor protection is weaker. These results suggest that ownership concentration is a key mechanism of corporate governance in such countries. Further, the identity of owners seems to matter for firm performance, as performance is higher when local institutions and domestic investors are involved.

Taken together, our results, obtained in the context of privatization, shed light on the functioning of corporate governance in emerging markets, particularly the role of ownership concentration and investor protection. Our results should also provide a new perspective for the ongoing debate on corporate governance reforms. Although

ownership concentration is an effective internal mechanism of corporate governance, it marginalizes (as suggested by several researchers, e.g., Holmström and Tirole, 1993) other external mechanisms of corporate governance (e.g., development of financial markets). Ownership concentration also impedes any effort to improve the efficiency of other institutions for corporate governance (e.g., legal institutions).

References

- Bekaert, G., Harvey C.R., Lundblad, C., 2001. Emerging equity markets and economic development. Journal of Development Economics 66, 465-504.
- Bortolotti, B., Fantini, M., Siniscalco, D., Vitalini, S., 2000. Privatization and institutions: A cross-country analysis, CESifo working paper series No. 375.
- Boubakri, N., Cosset, J.-C., 1998. The financial and operating performance of newly privatized firms: Evidence from developing countries. Journal of Finance 53, 1081-1110.
- Boutchkova, M. K., Megginson, W., 2000. The impact of privatization on capital market development and individual share ownership. Financial Management 29, 67-77.
- Boycko, M., Shleifer, A., Vishny, R., 1996. A theory of privatization, The Economic Journal 106, 309-319.
- Claessens, S., Djankov, S., 1999. Ownership concentration and corporate performance in the Czech Republic. Journal of Comparative Economics 27, 498-513.
- Demsetz, H., 1997. The firm in economic theory: A quiet revolution. American Economic Review 87, 426-429.
- Demsetz, H., Lehn, K., 1985. The structure of corporate ownership: causes and consequences. Journal of Political Economy 93, 1155-1177.
- Demsetz, H., Villalonga B., 2001. Ownership structure and corporate performance. Journal of Corporate Finance 7, 209-233.

- Denis, D.K., McConnell, J.J., 2003. International corporate governance, Journal of Financial and Quantitative Analysis 38, 1-36.
- Djankov, S., Murrell, P., 2002. Enterprise restructuring in transition: A quantitative survey.

 Journal of Economic Literature, 739-792.
- D'Souza, J., Megginson, W., 1999. The financial and operating performance of privatized firms during the 1990s. Journal of Finance 54, 1397-1438.
- Dyck, A., 2001. Privatization and corporate governance: principles, evidence, and future challenges. The World Bank Research Observer 16, 59-84.
- Dyck, A., Zingales, L., 2002. Private benefits of control: An international comparison. Journal of Finance, forthcoming.
- Frydman, R., Gray, C. W., Hessel, M., Rapaczynski, A., 1999. When does privatization work?

 The impact of private ownership on corporate performance in transition economies.

 Quarterly Journal of Economics 114, 1153-1191.
- Gupta, N., 2002. Partial privatization and firm performance, unpublished working paper, The William Davidson Institute, University of Michigan.
- Himmelberg, C.P., Hubbard, G., Palia, D., 1999. Understanding the determinants of managerial ownership and the link between ownership and performance. Journal of Financial Economics 53, 353-384.
- Holmström, B., Tirole, J., 1993. Market liquidity and performance monitoring. Journal of Political Economy 101, 678-709.

- La Porta, R., López-de-Silanes, F., Shleifer, A., Vishny, R., 1998, Law and Finance. Journal of Political Economy 106, 1113-1150.
- La Porta, R., López-de-Silanes, F., Shleifer, A., Vishny, R., 2000. Investor protection and corporate governance. Journal of Financial Economics 58, 3-27.
- McConnell, J., Servaes, H., 1990. Additional evidence on equity ownership and corporate value. Journal of Financial Economics 27, 595-612.
- Megginson, W., Nash, R., Netter, J.N., Poulsen, A.B., 2002. The choice of private versus public capital markets: Evidence from privatization. unpublished working paper, University of Oklahama, Wake Forest University and University of Georgia.
- Megginson, W., Nash, R., van Randenborgh, M., 1994. The financial and operating performance of newly privatized firms: An international empirical analysis. Journal of Finance 49, 403-452.
- Megginson, W., Netter, J.N., 2001 From state to market: A survey of empirical studies on privatization. Journal of Economic Literature 39, 321-389.
- Morck, R., Shleifer, A., Vishny, R., 1988. Management ownership and corporate performance: An empirical analysis. Journal of Financial Economics 20, 293-315.
- Palia, D., 2001. The Endogeneity of managerial compensation in firm valuation: A solution.

 Review of Financial Studies 14, 735-64.
- Perotti, E., 1995. Credible privatization. American Economic Review 85, 847-859.
- Perotti, E., Guney, S.E., 1993. Successful privatization plans: Enhanced credibility through timing and pricing of sales. Financial Management 22, 84-98.

- Perotti, E., van Oijen, P., 2001. Privatization, market development and political risk in emerging economies. Journal of International Money and Finance 20, 43-69.
- Shleifer, A., Vishny, R., 1997. A survey of corporate governance. Journal of Finance 52, 737-777.

Table 1 Description of the sample of newly privatized firms from emerging markets

This table provides some descriptive statistics for the sample of 170 privatized firms used in this study. We report the distribution of privatizations in the countries included in the sample by year, industry, legal origin, and region.

| | | | Distribution of privatizations | | |
|-------|---------|------------|---|--------|------------|
| | By year | | By industry | | |
| Year | Number | Percentage | Industry | Number | Percentage |
| 1980 | 1 | 0.59 | Financial | 57 | 33.53 |
| 1981 | 1 | 0.59 | Utilities | 19 | 11.18 |
| 1985 | 4 | 2.35 | Telecommunication | 13 | 7.65 |
| 1986 | 4 | 2.35 | Energy | 48 | 28.24 |
| 1987 | 3 | 1.76 | Other industries | 33 | 19.41 |
| 1988 | 3 | 1.76 | Total | 170 | 100 |
| 1989 | 23 | 13.53 | By Legal Origi | n | |
| 1990 | 14 | 8.24 | Category (countries) | Number | Percentage |
| 1991 | 26 | 15.29 | Common law countries (11) | 64 | 37.65 |
| 1992 | 20 | 11.76 | Civil law countries (15) | 106 | 62.35 |
| 1993 | 12 | 7.06 | Total (26) | 170 | 100 |
| 1994 | 17 | 10.00 | By Region * | | |
| 1995 | 7 | 4.12 | Region (countries) | Number | Percentage |
| 1996 | 16 | 9.41 | Africa and the Middle East (8) | 47 | 27.65 |
| 1997 | 19 | 11.18 | East and South Asia and the Pacific (8) | 46 | 27.06 |
| | | | Latin America and the Caribbean (8) | 55 | 32.35 |
| | | | Europe and Central Asia (2) | 22 | 12.94 |
| Total | 170 | 100 | Total (26) | 170 | 100 |

^{*} World Bank country group classifications.

Table 2
The evolution of ownership structure in newly privatized firms from emerging countries

This table presents summary statistics on the evolution of the ownership structure for a sample of 170 firms from 26 emerging countries privatized between 1980 and 1997. Panel A describes the ownership concentration measured by the percentage of shares held by the three largest investors (L3) and the Herfindahl index (H3). Panel B describes the evolution of ownership by type of investor. We consider five types of investors: the government, local institutions, foreign investors, employees, and individuals. The preprivatization ownership data (one year before privatization) come mainly from the offering prospectus, while the postprivatization data (year 0 to year 3) come from the annual reports and other additional sources such as Asian, Brazil and Mexico Company Handbooks, Worldscope Disclosure and The Guide to Asian Companies. All statistics are presented in percent. N refers to the number of observations.

| | Ownership share one year | | Ownership share (year relative | e after privatizat | |
|------------------------|--------------------------|---------------|--------------------------------|--------------------|-------|
| Type of investor | before privatization | 0 | +1 | +2 | +3 |
| | Panel A. Pri | vate ownersl | hip concentration | ٦ | |
| Cumulative share of th | e three largest inve | stors (L3) | | | |
| Mean | 10.64 | 38.57 | 42.85 | 45.55 | 45.96 |
| Median | 0.00 | 36.00 | 40.00 | 42.91 | 45.00 |
| N | 148 | 131 | 141 | 137 | 110 |
| Herfindahl index (H3) | | | | | |
| Mean | 3.36 | 20.58 | 21.87 | 23.12 | 23.42 |
| Median | 0.00 | 9.61 | 11.50 | 14.47 | 15.48 |
| N | 150 | 130 | 141 | 137 | 110 |
| | Pa | nel B. Type o | f investor | | |
| Government | | | | | |
| Mean | 78.16 | 35.23 | 30.22 | 25.02 | 21.41 |
| Median | 88.60 | 30.00 | 20.00 | 12.60 | 0.00 |
| N | 169 | 169 | 161 | 150 | 137 |
| Local Institutions | | | | | |
| Mean | 4.68 | 19.17 | 22.83 | 23.84 | 26.59 |
| Median | 0.00 | 1.25 | 8.83 | 10.00 | 8.61 |
| N | 120 | 109 | 124 | 123 | 98 |
| Foreign Investors | | | | | |
| Mean | 6.89 | 14.43 | 15.93 | 17.70 | 16.26 |
| Median | 0.00 | 0.00 | 2.20 | 4.96 | 3.90 |
| N | 137 | 136 | 142 | 136 | 117 |
| Employees | | | | | |
| Mean | 0.87 | 3.61 | 3.51 | 3.08 | 2.55 |
| Median | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 |
| N | 131 | 126 | 120 | 109 | 95 |
| Individuals | | | | | |
| Mean | 3.17 | 14.50 | 16.14 | 16.17 | 15.78 |
| Median | 0.00 | 8.60 | 10.00 | 9.50 | 8.47 |
| N | 118 | 111 | 118 | 112 | 90 |

Table 3
The evolution of the ownership structure in newly privatized firms from emerging markets classified by law origin

This table presents the evolution of the ownership structure for a sample of 170 firms from 26 emerging countries privatized between 1980 and 1997 according to the law origin. In Panel A, we focus on the ownership concentration measured by the percentage of shares held by the largest three investors (L3) and the Herfindahl index (H3). Panel B describes the evolution of ownership by type of investor: the government, local institutions, foreign investors, employees and individuals. We split the sample in two groups of firms, those from common law countries; and those from civil law countries. The preprivatization ownership data (one year before privatization) come mainly from the offering prospectus, while the postprivatization data (year 0 to year 3) come from the annual reports and other additional sources such as Asian, Brazil and Mexico Company Handbooks, Worldscope Disclosure and The guide to Asian Companies. The table also reports the p-value of the t-statistic for differences in means (p-value1) and medians (p-value2) between common law and civil law countries. All statistics are presented in percent. All tests have been conducted with at least 30 observations.

| | Ownership share one year | 0 | wnership share (year relative t | after privatization) | on |
|------------------------|--------------------------|------------------|------------------------------------|----------------------|--------|
| Type of investor | before privatization | 0 | +1 | +2 | +3 |
| | Panel A. Priva | ite ownership | concentration | | |
| Cumulative share of th | e three largest investo | ors (L3) | | | |
| Common law countries | 5 | | | | |
| Mean | 13.97 | 22.67 | 26.07 | 28.09 | 28.64 |
| Median | 0.00 | 23.75 | 31.19 | 32.00 | 31.69 |
| Civil law countries | | | | | |
| Mean | 8.29 | 49.03 | 54.57 | 56.57 | 58.42 |
| Median | 0.00 | 52.00 | 55.43 | 59.00 | 60.20 |
| Means difference | 5.68 | -26.36 | -28.49 | -28.48 | -29.78 |
| P-value 1 | 0.072 | 0.000 | 0.000 | 0.000 | 0.000 |
| P-value 2 | 0.114 | 0.000 | 0.000 | 0.000 | 0.000 |
| Herfindahl index (H3) | | | | | |
| Common law countries | S | | | | |
| Mean | 4.55 | 6.95 | 7.64 | 8.49 | 8.39 |
| Median | 0.00 | 3.02 | 5.23 | 6.27 | 6.11 |
| Civil law countries | | | | | |
| Mean | 2.54 | 29.37 | 31.81 | 32.36 | 34.22 |
| Median | 0.00 | 17.21 | 26.00 | 25.50 | 26.96 |
| Means difference | 2.02 | -22.43 | -24.17 | -23.86 | -25.83 |
| P-value 1 | 0.085 | 0.000 | 0.000 | 0.000 | 0.000 |
| P-value 2 | 0.095 | 0.000 | 0.000 | 0.000 | 0.000 |
| | Pane | el B. Type of in | vestor | | |
| Government | | | | | |
| Common law countries | | | | | |
| Mean | 77.00 | 37.91 | 37.42 | 33.09 | 30.82 |
| Median | 79.50 | 35.47 | 38.91 | 30.25 | 30.00 |
| Civil law countries | | | | | |
| Mean | 78.86 | 33.60 | 25.72 | 19.94 | 15.65 |
| Median | 89.10 | 28.60 | 12.96 | 0.00 | 0.00 |
| Means difference | -1.86 | 4.31 | 11.70 | 13.15 | 15.17 |
| P-value 1 | 0.632 | 0.393 | 0.020 | 0.008 | 0.002 |
| P-value 2 | 0.749 | 0.345 | 0.019 | 0.003 | 0.001 |

Table 3 (continued)

| | Ownership share one year _ | 0 | wnership share (year relative t | after privatization) | on |
|----------------------|----------------------------|--------|------------------------------------|----------------------|--------|
| Type of investor | before privatization | 0 | +1 | +2 | +3 |
| Local Institutions | ' | | | | |
| Common law countries | | | | | |
| Mean | 4.18 | 8.38 | 10.82 | 11.84 | 13.56 |
| Median | 0.00 | 0.00 | 3.11 | 3.78 | 3.11 |
| Civil law countries | | | | | |
| Mean | 5.04 | 26.48 | 30.94 | 31.26 | 35.58 |
| Median | 0.00 | 6.20 | 17.76 | 16.00 | 23.16 |
| Means difference | -0.86 | -18.10 | -20.12 | -19.42 | -22.03 |
| P-value 1 | 0.705 | 0.001 | 0.000 | 0.000 | 0.001 |
| P-value 2 | 0.634 | 0.007 | 0.004 | 0.005 | 0.015 |
| Foreign investors | | | | | |
| Common law countries | | | | | |
| Mean | 13.64 | 15.08 | 15.97 | 16.53 | 16.05 |
| Median | 0.00 | 1.73 | 5.95 | 7.65 | 9.02 |
| Civil law countries | 0.00 | 1.75 | 3.73 | 7.05 | 7.02 |
| Mean | 2.37 | 14.04 | 15.91 | 18.38 | 16.40 |
| Median | 0.00 | 0.00 | 0.54 | 0.96 | 0.54 |
| Median | 0.00 | 0.00 | 0.54 | 0.76 | 0.54 |
| Means difference | 11.27 | 1.04 | 0.06 | -1.85 | -0.35 |
| P-value 1 | 0.000 | 0.784 | 0.987 | 0.656 | 0.935 |
| P-value 2 | 0.000 | 0.312 | 0.330 | 0.410 | 0.231 |
| Employees | | | | | |
| Common law countries | | | | | |
| Mean | 1.12 | 3.70 | 3.61 | 3.07 | 3.21 |
| Median | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 |
| Civil law countries | | | | | |
| Mean | 0.72 | 3.55 | 3.45 | 3.09 | 2.23 |
| Median | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 |
| Means difference | 0.40 | 0.15 | 0.16 | -0.02 | 0.97 |
| P-value 1 | 0.530 | 0.938 | 0.939 | 0.993 | 0.657 |
| P-value 2 | 0.082 | 0.074 | 0.058 | 0.093 | 0.269 |
| Individuals | | | | | |
| Common law countries | | | | | |
| Mean | 4.39 | 27.38 | 28.59 | 29.56 | 30.39 |
| Median | 0.00 | 19.30 | 22.00 | 22.50 | 23.91 |
| Civil law countries | 0.44 | 7.50 | 0.01 | 0.01 | 0.11 |
| Mean | 2.44 | 7.53 | 9.01 | 9.01 | 8.11 |
| Median | 0.00 | 0.00 | 4.00 | 1.74 | 0.00 |
| Means difference | 1.95 | 19.85 | 19.59 | 20.55 | 22.28 |
| P-value 1 | 0.229 | 0.000 | 0.000 | 0.000 | 0.000 |
| P-value 2 | 0.575 | 0.000 | 0.000 | 0.000 | 0.000 |

Table 4 Summary of the variables

This table describes the variables used in our regression analysis to investigate the determinants of the postprivatization private ownership concentration.

| Variables | Definition |
|-----------|---|
| | Panel A. Firm-level variables |
| RISK | The standard deviation of the annual return on equity ratios. |
| SALESGR | Sales growth. |
| SIZE | The natural logarithm of total sales. |
| IND1 | A dummy variable equal to unity if the firm is a bank or insurance company. |
| IND2 | A dummy variable equal to unity if the firm is a utility. |
| IND3 | A dummy variable equal to unity if the firm is a telecommunication company. |
| IND4 | A dummy variable equal to unity if the firm belongs to oil and gas, petrol/petrochemical, cement and mining. |
| IND5 | A dummy variable equal to unity if the firm belongs to other industries. |
| LATE | A dummy variable equal to unity if the sample firm is privatized after the median privatization date in the country and zero otherwise. |
| | Panel B. Country-level variables |
| LAW | A dummy variable equal to unity for firms from common law countries. |
| JUDICIAL | La Porta's (1998) index of the efficiency of the judicial system. |
| GINI | The United Nation's index of income inequality. |
| GS | The ICRG assessment of a country's government stability. |
| ET | The ICRG assessment of a country's ethnic tensions. |

Table 5
The determinants of private ownership concentration

The table shows the regression results of private ownership concentration measured by the logistic transformation of the percentage held by the three major shareholders (*LL3*) on the set of explanatory variables indicated below. *REGION1*, *REGION2*, *REGION3* and *REGION4* refer to Africa, Latin America, Asia, and Europe, respectively. *VAR* refers to ONE (unit vector) (Specification (1)), *GINI* (specification (2)), *GS* (specification (3)), and *ET* (specification (4)). All regressions include year-fixed effects (coefficients' estimates not reported). Heteroskedasticity robust p-values are in parentheses. A constant term is included in each regression. *, ** and *** Significant at the 10 percent, 5 percent and 1 percent level, respectively. Variable definitions for the acronyms are reported in Table 4.

| Variable | (1) | (2) | (3) | (4) |
|-------------------------|-----------|-----------|-----------|-----------|
| RISK | 0.183 | 0.174 | 0.216 | 0.172 |
| | (0.148) | (0.176) | (0.076) | (0.175) |
| SALESGR | 0.380*** | 0.387*** | 0.406*** | 0.418*** |
| | (0.000) | (0.000) | (0.000) | (<.0001) |
| SIZE | -0.457*** | -0.478*** | -0.444*** | -0.424*** |
| | (<.0001) | (<.0001) | (<.0001) | (<.0001) |
| IND1 | 1.710*** | 1.783*** | 1.730*** | 1.594*** |
| | (<.0001) | (<.0001) | (<.0001) | (<.0001) |
| IND2 | -1.626*** | -1.710*** | -1.671*** | -1.617*** |
| | (<.0001) | (<.0001) | (<.0001) | (<.0001) |
| IND3 | -0.853** | -0.827* | -0.877** | -0.801* |
| | (0.056) | (0.062) | (0.044) | (0.082) |
| IND4 | 0.271 | 0.237 | 0.264 | 0.271 |
| | (0.317) | (0.376) | (0.330) | (0.316) |
| IND5 | 0.498** | 0.517** | 0.554** | 0.553** |
| | (0.043) | (0.036) | (0.026) | (0.023) |
| LATE | -0.295 | -0.278 | -0.546 | -0.213 |
| | (0.394) | (0.423) | (0.113) | (0.528) |
| LAW | -0.969** | -0.963** | -1.697*** | -1.048** |
| | (0.044) | (0.040) | (0.000) | (0.023) |
| JUDICIAL | 0.007 | 0.017 | 0.066 | -0.021 |
| | (0.942) | (0.851) | (0.481) | (0.813) |
| GINI | 0.024 | , , | 0.034 | 0.016 |
| | (0.290) | | (0.114) | (0.501) |
| GS | 0.161** | 0.152** | , , | 0.192** |
| | (0.031) | (0.040) | | (0.011) |
| ET | 0.352*** | 0.341*** | 0.436*** | , , |
| | (0.001) | (0.001) | (<.0001) | |
| VAR*REGION1 | -1.692*** | -0.016 | -0.093 | 0.050 |
| | (<.0001) | (0.470) | (0.326) | (0.710) |
| VAR*REGION2 | 2.412*** | 0.071*** | 0.368*** | 0.933*** |
| | (<.0001) | (<.0001) | (<.0001) | (<.0001) |
| VAR*REGION3 | -0.716** | 0.009 | 0.054 | 0.189* |
| | (0.016) | (0.691) | (0.446) | (0.087) |
| VAR*REGION4 | -0.005 | 0.027 | 0.066 | 0.394*** |
| | (0.988) | (0.226) | (0.505) | (0.001) |
| Adjusted R ² | 0.497 | 0.496 | 0.491 | 0.501 |
| N | 402 | 402 | 402 | 402 |

Table 6
The performance of newly privatized firms

This table presents descriptive statistics on the performance of a sample of 170 firms in 26 countries privatized between 1980 and 1997. The performance measures are return on sales (ROS), return on assets (ROA), and return on equity (ROE). We report the mean and median levels of performance from three years after privatization, and the mean and median changes in performance from before to after privatization. Panel A and Panel B report the unadjusted and market-adjusted performance measures, respectively. *** Significant at the 1 percent level.

| | | | | Levels | | | | Cho | Changes |
|-------------|----------------------|------------|---------------|----------|-----------|---------------------|----------|----------------|----------------|
| | Before privatization | vatization | privatization | | After pri | After privatization | | | |
| | | One Year | | | | | | | Wilcoxon Two |
| Performance | Before | before | Year 0 | One year | Two years | Three years | After | | Sample Test Z- |
| measures | [-1,-3] | -1 | 0 | +1 | +2 | +3 | [+1, +3] | [-1,-3; +1,+3] | Statistic |
| ROS | | | | | | | | | |
| Mean | 0.075 | 0.099 | 0.128 | 0.162 | 0.159 | 0.062 | 0.152 | 0.077 | |
| Median | 0.050 | 0.064 | 0.096 | 0.100 | 0.094 | 0.092 | 0.109 | 0.039 | 4.531*** |
| Z | 164 | 170 | 166 | 170 | 170 | 136 | 164 | 164 | |
| ROA | | | | | | | | | |
| Mean | 0.043 | 0.051 | 0.055 | 0.063 | 0.061 | 0.060 | 0.067 | 0.024 | |
| Median | 0.024 | 0.026 | 0.041 | 0.044 | 0.043 | 0.042 | 0.047 | 0.011 | 3.759*** |
| Z | 165 | 171 | 166 | 171 | 172 | 137 | 165 | 165 | |
| ROE | | | | | | | | | |
| Mean | 0.163 | 0.224 | 0.187 | 0.197 | 0.220 | 0.214 | 0.224 | 0.061 | |
| Median | 0.096 | 0.109 | 0.127 | 0.145 | 0.131 | 0.140 | 0.145 | 0.037 | 3.243*** |
| Z | 165 | 173 | 168 | 171 | 172 | 138 | 165 | 165 | |

2SLS estimation of the effect of ownership structure on the performance of newly privatized firms (second stage results)

logarithm of total sales. FREEDOM controls for a country's level of economic and institutional development and GDPG controls for the economic growth. PROTEC is the ICRG's measure of legal protection. LINS, FOREIGN, EMPLOYEE, and INDIVIDUALS are the percentages held by local institutions, foreign investors, employees, The table reports the 2SLS regression results of the relation between ownership structure and firm performance. Panel A examines the role of private ownership concentration. Panel B examines the role of ownership identity. We measure firm performance by the return on sales (ROS). LCONC is the ownership concentration and individuals, respectively. All regressions include year-fixed effects (coefficients' estimates not reported). Heteroskedasticity robust p-values appear in term in the performance equation and Z_{i} are the instrumental variables. *, ** and *** Significant at the 10 percent, 5 percent and 1 percent level, respectively. parentheses. N refers to the number of firm-year observations. The last column reports the F-statistic and p-value for the test of $E[\epsilon_{1i}, Z_{ti}] = 0$; where ϵ_{1i} is the error measured by the logistic transformation of the percentage held by the three major shareholders (LL3). LCONC2 is the squared value of LCONC. SIZE is the natural

| (i) | | | (iii) | ≘ | (i) | | |
|-------------------------------------|--------------------------|---|---------------------|--------------------------|---------------------|--------------------------|--|
| -0.870 (0.103) | INTERCEPT | | 0.216 (0.011) | 0.208 (0.018) | 0.103 (0.242) | INTERCEPT | |
| 2.003* (0.057) | LINS | | 0.072*** (0.007) | 0.111*** | 0.022*** (0.002) | LCONC | |
| -1.478 (0.152) | FOREIGN | | -0.018** (0.019) | -0.031*** (<.0001) | | LCONC*PROTEC | |
| 0.986 (0.315) | EMPLOYEE | | 0.009*** | | | C LCONC2 | Pa |
| 3.028* (0.051) | employee individuals | Panel B. The | -0.017** (0.012) | -0.012* (0.063) | -0.017** (0.013) | SIZE | nel A. The ro |
| 0.006 (0.714) | SIZE | e role of own | 0.035*** (0.008) | 0.039*** | 0.047*** | FREEDOM | le of owners |
| 0.053 (0.105) | FREEDOM | Panel B. The role of ownership identity | 0.002 (0.531) | 0.004 | 0.007* (0.073) | GDPG | Panel A. The role of ownership concentration |
| 0.001 (0.882) | GDPG | Ϋ | -0.027** (0.027) | -0.040*** (0.002) | | PROTEC | ation |
| Included | Industry | | Included | Included | Included | Industry | |
| 0.067 | Adjusted R ² | | 0.222 | 0.195 | 0.098 | Adjusted R ² | |
| 218 | z | | 415 | 415 | 415 | z | |
| 0.08 <i>4</i> * [0.08 <i>4</i>] | F-statistic [p-value] | | 0.80 [0.586] | 0. <i>7</i> 9 [0.599] | 1.39 [0.207] | F-statistic [p-value] | |

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our working papers are available on the Internet at the following addresses:

http://www.feem.it/web/activ/_wp.html http://papers.ssrn.com

| SUST | 1.2002 | K. TANO, M.D. FAMINOW, M. KAMUANGA and B. SWALLOW: Using Conjoint Analysis to Estimate Farmers' |
|-----------|---------|--|
| | | Preferences for Cattle Traits in West Africa |
| ETA | 2.2002 | Efrem CASTELNUOVO and Paolo SURICO: What Does Monetary Policy Reveal about Central Bank's |
| | | Preferences? |
| WAT | 3.2002 | Duncan KNOWLER and Edward BARBIER: The Economics of a "Mixed Blessing" Effect: A Case Study of the |
| | | Black Sea |
| CLIM | 4.2002 | Andreas LÖSCHEL: Technological Change in Economic Models of Environmental Policy: A Survey |
| VOL | 5.2002 | Carlo CARRARO and Carmen MARCHIORI: Stable Coalitions |
| CLIM | 6.2002 | Marzio GALEOTTI, Alessandro LANZA and Matteo MANERA: Rockets and Feathers Revisited: An International |
| | ***** | Comparison on European Gasoline Markets |
| ETA | 7.2002 | Effrosyni DIAMANTOUDI and Eftichios S. SARTZETAKIS: Stable International Environmental Agreements: An |
| | | Analytical Approach |
| KNOW | 8.2002 | Alain DESDOIGTS: Neoclassical Convergence Versus Technological Catch-up: A Contribution for Reaching a |
| | | Consensus |
| NRM | 9.2002 | Giuseppe DI VITA: Renewable Resources and Waste Recycling |
| KNOW | 10.2002 | Giorgio BRUNELLO: Is Training More Frequent when Wage Compression is Higher? Evidence from 11 |
| | | European Countries |
| ETA | 11.2002 | Mordecai KURZ, Hehui JIN and Maurizio MOTOLESE: Endogenous Fluctuations and the Role of Monetary |
| | | <u>Policy</u> |
| KNOW | 12.2002 | Reyer GERLAGH and Marjan W. HOFKES: Escaping Lock-in: The Scope for a Transition towards Sustainable |
| | 12.200 | Growth? |
| NRM | 13.2002 | Michele MORETTO and Paolo ROSATO: The Use of Common Property Resources: A Dynamic Model |
| CLIM | 14.2002 | Philippe QUIRION: Macroeconomic Effects of an Energy Saving Policy in the Public Sector |
| CLIM | 15.2002 | Roberto ROSON: Dynamic and Distributional Effects of Environmental Revenue Recycling Schemes: |
| CLIM | 16.2002 | Simulations with a General Equilibrium Model of the Italian Economy Francesco RICCI (1): Environmental Policy Growth when Inputs are Differentiated in Pollution Intensity |
| ETA | 17.2002 | Alberto PETRUCCI: Devaluation (Levels versus Rates) and Balance of Payments in a Cash-in-Advance |
| EIA | 17.2002 | Economy |
| Coalition | 18.2002 | |
| Theory | 10.2002 | László Á. KÓCZY (liv): The Core in the Presence of Externalities |
| Network | | |
| Coalition | 19.2002 | Steven J. BRAMS, Michael A. JONES and D. Marc KILGOUR (liv): Single-Peakedness and Disconnected |
| Theory | | · / · · · · |
| Network | | Coalitions |
| Coalition | 20.2002 | Guillaume HAERINGER (liv): On the Stability of Cooperation Structures |
| Theory | | ommume Inizitivozit (iiv). On the stability of cooperation structures |
| Network | | |
| NRM | 21.2002 | Fausto CAVALLARO and Luigi CIRAOLO: Economic and Environmental Sustainability: A Dynamic Approach |
| | | in Insular Systems |
| CLIM | 22.2002 | Barbara BUCHNER, Carlo CARRARO, Igor CERSOSIMO and Carmen MARCHIORI: Back to Kyoto? US |
| CL D.4 | 22 2002 | Participation and the Linkage between R&D and Climate Cooperation |
| CLIM | 23.2002 | Andreas LÖSCHEL and ZhongXIANG ZHANG: The Economic and Environmental Implications of the US |
| ETA | 24.2002 | Repudiation of the Kyoto Protocol and the Subsequent Deals in Bonn and Marrakech Marzio GALEOTTI, Louis J. MACCINI and Fabio SCHIANTARELLI: Inventories, Employment and Hours |
| CLIM | 25.2002 | Hannes EGLI: Are Cross-Country Studies of the Environmental Kuznets Curve Misleading? New Evidence from |
| CLIM | 23.2002 | Time Series Data for Germany |
| ETA | 26.2002 | Adam B. JAFFE, Richard G. NEWELL and Robert N. STAVINS: Environmental Policy and Technological |
| 1.111 | 20.2002 | Change |
| SUST | 27.2002 | Joseph C. COOPER and Giovanni SIGNORELLO: Farmer Premiums for the Voluntary Adoption of |
| | | Conservation Plans |
| SUST | 28.2002 | The ANSEA Network: Towards An Analytical Strategic Environmental Assessment |
| KNOW | 29.2002 | Paolo SURICO: Geographic Concentration and Increasing Returns: a Survey of Evidence |
| ETA | 30.2002 | Robert N. STAVINS: Lessons from the American Experiment with Market-Based Environmental Policies |

| NRM | 21 2002 | |
|--|--|---|
| | 31.2002 | Carlo GIUPPONI and Paolo ROSATO: Multi-Criteria Analysis and Decision-Support for Water Management at |
| | | the Catchment Scale: An Application to Diffuse Pollution Control in the Venice Lagoon |
| NRM | 32.2002 | Robert N. STAVINS: National Environmental Policy During the Clinton Years |
| KNOW | 33.2002 | A. SOUBEYRAN and H. STAHN: Do Investments in Specialized Knowledge Lead to Composite Good |
| KNOW | 24.2002 | Industries? |
| KNOW | 34.2002 | G. BRUNELLO, M.L. PARISI and Daniela SONEDDA: <u>Labor Taxes</u> , Wage Setting and the Relative Wage |
| CL D.4 | 25 2002 | Effect |
| CLIM | 35.2002 | C. BOEMARE and P. QUIRION (lv): Implementing Greenhouse Gas Trading in Europe: Lessons from |
| CL D.4 | 26.2002 | Economic Theory and International Experiences |
| CLIM | 36.2002 | T.TIETENBERG (IV): The Tradable Permits Approach to Protecting the Commons: What Have We Learned? |
| CLIM | 37.2002 | K. REHDANZ and R.J.S. TOL (IV): On National and International Trade in Greenhouse Gas Emission Permits |
| CLIM | 38.2002 | C. FISCHER (IV): Multinational Taxation and International Emissions Trading |
| SUST | 39.2002 | G. SIGNORELLO and G. PAPPALARDO: Farm Animal Biodiversity Conservation Activities in Europe under |
| NRM | 40.2002 | the Framework of Agenda 2000 S.M. CAVANAGH, W. M. HANEMANN and R. N. STAVINS: Muffled Price Signals: Household Water Demand |
| INIXIVI | 40.2002 | under Increasing-Block Prices |
| NRM | 41.2002 | A. J. PLANTINGA, R. N. LUBOWSKI and R. N. STAVINS: The Effects of Potential Land Development on |
| INIXIVI | 41.2002 | Agricultural Land Prices |
| CLIM | 42,2002 | C. OHL (Ivi): Inducing Environmental Co-operation by the Design of Emission Permits |
| CLIM | 43.2002 | J. EYCKMANS, D. VAN REGEMORTER and V. VAN STEENBERGHE (Ivi): Is Kyoto Fatally Flawed? An |
| CLIM | 43.2002 | Analysis with MacGEM |
| CLIM | 44.2002 | A. ANTOCI and S. BORGHESI (Ivi): Working Too Much in a Polluted World: A North-South Evolutionary |
| CLIM | 77.2002 | Model |
| ETA | 45.2002 | P. G. FREDRIKSSON, Johan A. LIST and Daniel MILLIMET (Ivi): Chasing the Smokestack: Strategic |
| LIII | 13.2002 | Policymaking with Multiple Instruments |
| ETA | 46.2002 | Z. YU (Ivi): A Theory of Strategic Vertical DFI and the Missing Pollution-Haven Effect |
| SUST | 47.2002 | Y. H. FARZIN: Can an Exhaustible Resource Economy Be Sustainable? |
| SUST | 48.2002 | Y. H. FARZIN: Sustainability and Hamiltonian Value |
| | | |
| KNOW | 49.2002 | C. PIGA and M. VIVARELLI: Cooperation in R&D and Sample Selection |
| Coalition | 50.2002 | M. SERTEL and A. SLINKO (liv): Ranking Committees, Words or Multisets |
| Theory | | |
| Network | | |
| Coalition | 51.2002 | Sergio CURRARINI (liv): Stable Organizations with Externalities |
| Theory | | |
| Network | | |
| ETA | 52.2002 | Robert N. STAVINS: Experience with Market-Based Policy Instruments |
| ETA | 53.2002 | |
| | 33.2002 | |
| LIA | | C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and P. KLEIN (199): Integrated Assessment Modeling: Modules for Cooperation |
| | | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation |
| CLIM | 54.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty |
| | | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- |
| CLIM ETA | 54.2002 55.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies |
| CLIM ETA SUST | 54.2002 55.2002 56.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs |
| CLIM ETA | 54.2002 55.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of |
| CLIM ETA SUST SUST | 54.2002 55.2002 56.2002 57.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests |
| CLIM ETA SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy |
| CLIM ETA SUST SUST | 54.2002 55.2002 56.2002 57.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions |
| CLIM ETA SUST SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? |
| CLIM ETA SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together |
| CLIM ETA SUST SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? |
| CLIM ETA SUST SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together |
| CLIM ETA SUST SUST SUST SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union |
| CLIM ETA SUST SUST SUST SUST VOL | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic |
| CLIM ETA SUST SUST SUST VOL | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity |
| CLIM ETA SUST SUST SUST SUST VOL ETA PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability |
| CLIM ETA SUST SUST SUST VOL | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q |
| CLIM ETA SUST SUST SUST SUST VOL ETA PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edit DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edit DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 66.2002 67.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITIMA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 66.2002 67.2002 68.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM CLIM | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 68.2002 68.2002 69.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 66.2002 67.2002 68.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? Anna Alberini, Patrizia RIGANTI and Alberto LONGO: Can People Value the Aesthetic and Use Services of |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM CLIM SUST | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 68.2002 68.2002 69.2002 70.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edin DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? Anna AlbERINI, Patrizia RIGANTI and Alberto LONGO: Can People Value the Aesthetic and Use Services of Urban Sites? Evidence from a Survey of Belfast Residents |
| CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM CLIM | 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 68.2002 68.2002 69.2002 | R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? Anna Alberini, Patrizia RIGANTI and Alberto LONGO: Can People Value the Aesthetic and Use Services of |

| | | |
|---------|----------|--|
| NRM | 72.2002 | Philippe BONTEMS and Pascal FAVARD: Input Use and Capacity Constraint under Uncertainty: The Case of |
| DD IV | 72 2002 | Irrigation Mel ground OMP AN: The Performance of State Owned Enterprises and Newly Privatived Firms: Empirical |
| PRIV | 73.2002 | Mohammed OMRAN: The Performance of State-Owned Enterprises and Newly Privatized Firms: Empirical Evidence from Egypt |
| PRIV | 74.2002 | Mike BURKART, Fausto PANUNZI and Andrei SHLEIFER: Family Firms |
| PRIV | 75.2002 | Emmanuelle AURIOL, Pierre M. PICARD: Privatizations in Developing Countries and the Government Budget |
| 1111 | 70.2002 | Constraint |
| PRIV | 76.2002 | Nichole M. CASTATER: Privatization as a Means to Societal Transformation: An Empirical Study of |
| | | Privatization in Central and Eastern Europe and the Former Soviet Union |
| PRIV | 77.2002 | Christoph LÜLSFESMANN: Benevolent Government, Managerial Incentives, and the Virtues of Privatization |
| PRIV | 78.2002 | Kate BISHOP, Igor FILATOTCHEV and Tomasz MICKIEWICZ: Endogenous Ownership Structure: Factors |
| | | Affecting the Post-Privatisation Equity in Largest Hungarian Firms |
| PRIV | 79.2002 | Theodora WELCH and Rick MOLZ: How Does Trade Sale Privatization Work? |
| DD II I | 00.2002 | Evidence from the Fixed-Line Telecommunications Sector in Developing Economies |
| PRIV | 80.2002 | Alberto R. PETRUCCI: Government Debt, Agent Heterogeneity and Wealth Displacement in a Small Open Economy |
| CLIM | 81.2002 | Timothy SWANSON and Robin MASON (lvi): The Impact of International Environmental Agreements: The Case |
| CLIM | 01.2002 | of the Montreal Protocol |
| PRIV | 82.2002 | George R.G. CLARKE and Lixin Colin XU: Privatization, Competition and Corruption: How Characteristics of |
| | | Bribe Takers and Payers Affect Bribe Payments to Utilities |
| PRIV | 83.2002 | Massimo FLORIO and Katiuscia MANZONI: The Abnormal Returns of UK Privatisations: From Underpricing |
| | | to Outperformance |
| NRM | 84.2002 | Nelson LOURENÇO, Carlos RUSSO MACHADO, Maria do ROSÁRIO JORGE and Luís RODRIGUES: <u>An</u> |
| | | Integrated Approach to Understand Territory Dynamics. The Coastal Alentejo (Portugal) |
| CLIM | 85.2002 | Peter ZAPFEL and Matti VAINIO (Iv): Pathways to European Greenhouse Gas Emissions Trading History and |
| CI D I | 06.2002 | Misconceptions Response of the second of th |
| CLIM | 86.2002 | Pierre COURTOIS: Influence Processes in Climate Change Negotiations: Modelling the Rounds |
| ETA | 87.2002 | Vito FRAGNELLI and Maria Erminia MARINA (Iviii): Environmental Pollution Risk and Insurance |
| ETA | 88.2002 | Laurent FRANCKX (Iviii): Environmental Enforcement with Endogenous Ambient Monitoring Timo GOESCHL and Timothy M. SWANSON (Iviii): Lost Horizons. The noncooperative management of an |
| ETA | 89.2002 | evolutionary biological system. |
| ETA | 90.2002 | Hans KEIDING (Iviii): Environmental Effects of Consumption: An Approach Using DEA and Cost Sharing |
| ETA | 91.2002 | Wietze LISE (Iviii): A Game Model of People's Participation in Forest Management in Northern India |
| CLIM | 92.2002 | Jens HORBACH: Structural Change and Environmental Kuznets Curves |
| ETA | 93.2002 | Martin P. GROSSKOPF: Towards a More Appropriate Method for Determining the Optimal Scale of Production |
| DIII | 75.2002 | Units |
| VOL | 94.2002 | Scott BARRETT and Robert STAVINS: Increasing Participation and Compliance in International Climate Change |
| | | Agreements |
| CLIM | 95.2002 | Banu BAYRAMOGLU LISE and Wietze LISE: Climate Change, Environmental NGOs and Public Awareness in |
| | | the Netherlands: Perceptions and Reality |
| CLIM | 96.2002 | Matthieu GLACHANT: The Political Economy of Emission Tax Design in Environmental Policy |
| KNOW | 97.2002 | Kenn ARIGA and Giorgio BRUNELLO: Are the More Educated Receiving More Training? Evidence from |
| ET A | 00.2002 | Thailand Girling FORTE AND MANERA FOR SIGNAL STATE OF THE STATE OF TH |
| ETA | 98.2002 | Ganfranco FORTE and Matteo MANERA: Forecasting Volatility in European Stock Markets with Non-linear |
| ETA | 99.2002 | GARCH Models Geoffrey HEAL: Bundling Biodiversity |
| ETA | 100.2002 | Geoffrey HEAL, Brian WALKER, Simon LEVIN, Kenneth ARROW, Partha DASGUPTA, Gretchen DAILY, Paul |
| LIM | 100.2002 | EHRLICH, Karl-Goran MALER, Nils KAUTSKY, Jane LUBCHENCO, Steve SCHNEIDER and David |
| | | STARRETT: Genetic Diversity and Interdependent Crop Choices in Agriculture |
| ETA | 101.2002 | Geoffrey HEAL: Biodiversity and Globalization |
| VOL | 102.2002 | Andreas LANGE: Heterogeneous International Agreements – If per capita emission levels matter |
| ETA | 103.2002 | Pierre-André JOUVET and Walid OUESLATI: Tax Reform and Public Spending Trade-offs in an Endogenous |
| | | Growth Model with Environmental Externality |
| ETA | 104.2002 | Anna BOTTASSO and Alessandro SEMBENELLI: Does Ownership Affect Firms' Efficiency? Panel Data |
| | | Evidence on Italy |
| PRIV | 105.2002 | Bernardo BORTOLOTTI, Frank DE JONG, Giovanna NICODANO and Ibolya SCHINDELE: Privatization and |
| | | Stock Market Liquidity |
| ETA | 106.2002 | Haruo IMAI and Mayumi HORIE (Iviii): Pre-Negotiation for an International Emission Reduction Game |
| PRIV | 107.2002 | Sudeshna GHOSH BANERJEE and Michael C. MUNGER: Move to Markets? An Empirical Analysis of |
| DDIV | 100 2002 | Privatisation in Developing Countries Civilianna CIPMENS and Mishal CIVII APD: Privatization and Investment: Crawding Out Effect vs Financial |
| PRIV | 108.2002 | Guillaume GIRMENS and Michel GUILLARD: Privatization and Investment: Crowding-Out Effect vs Financial |
| PRIV | 109.2002 | <u>Diversification</u> Alberto CHONG and Florencio LÓPEZ-DE-SILANES: Privatization and Labor Force Restructuring Around the |
| 1 1/1 / | 107.2002 | World |
| PRIV | 110.2002 | Nandini GUPTA: Partial Privatization and Firm Performance |
| PRIV | 111.2002 | François DEGEORGE, Dirk JENTER, Alberto MOEL and Peter TUFANO: Selling Company Shares to |
| | 002 | Reluctant Employees: France Telecom's Experience |
| | | |

| PRIV | 112.2002 | Isaac OTCHERE: Intra-Industry Effects of Privatization Announcements: Evidence from Developed and |
|-------------|------------------|--|
| | | <u>Developing Countries</u> |
| PRIV | 113.2002 | Yannis KATSOULAKOS and Elissavet LIKOYANNI: Fiscal and Other Macroeconomic Effects of Privatization |
| PRIV | 114.2002 | Guillaume GIRMENS: Privatization, International Asset Trade and Financial Markets |
| PRIV | 115.2002 | D. Teja FLOTHO: A Note on Consumption Correlations and European Financial Integration |
| PRIV | 116.2002 | Ibolya SCHINDELE and Enrico C. PEROTTI: Pricing Initial Public Offerings in Premature Capital Markets: |
| PRIV | 1.2003 | The Case of Hungary Gabriella CHIESA and Giovanna NICODANO: Privatization and Financial Market Development: Theoretical |
| DD III | 2 2002 | Issues One of the control of the co |
| PRIV | 2.2003 | Ibolya SCHINDELE: Theory of Privatization in Eastern Europe: Literature Review |
| PRIV | 3.2003 | Wietze LISE, Claudia KEMFERT and Richard S.J. TOL: Strategic Action in the Liberalised German Electricity |
| Cr. D. f | 4.2002 | Market Market |
| CLIM | 4.2003 | Laura MARSILIANI and Thomas I. RENSTRÖM: Environmental Policy and Capital Movements: The Role of |
| KNOW | 5 2002 | Government Commitment |
| KNOW | 5.2003 | Reyer GERLAGH: Induced Technological Change under Technological Competition |
| ETA | 6.2003 | Efrem CASTELNUOVO: Squeezing the Interest Rate Smoothing Weight with a Hybrid Expectations Model |
| SIEV | 7.2003 | Anna ALBERINI, Alberto LONGO, Stefania TONIN, Francesco TROMBETTA and Margherita TURVANI: The |
| | | Role of Liability, Regulation and Economic Incentives in Brownfield Remediation and Redevelopment: |
| NDM | 9 2002 | Evidence from Surveys of Developers Elissaios PAPYRAKIS and Reyer GERLAGH: Natural Resources: A Blessing or a Curse? |
| NRM CLIM | 8.2003 9.2003 | A. CAPARRÓS, JC. PEREAU and T. TAZDAÏT: North-South Climate Change Negotiations: a Sequential Game |
| KNOW | 10.2002 | with Asymmetric Information |
| KNOW | 10.2003 | Giorgio BRUNELLO and Daniele CHECCHI: School Quality and Family Background in Italy |
| CLIM | 11.2003 | Efrem CASTELNUOVO and Marzio GALEOTTI: Learning By Doing vs Learning By Researching in a Model of |
| KNOW | 12 2002 | Climate Change Policy Analysis |
| KNOW | 12.2003 | Carole MAIGNAN, Gianmarco OTTAVIANO and Dino PINELLI (eds.): Economic Growth, Innovation, Cultural |
| KNOW | 12 2002 | Diversity: What are we all talking about? A critical survey of the state-of-the-art |
| KNOW | 13.2003 | Carole MAIGNAN, Gianmarco OTTAVIANO, Dino PINELLI and Francesco RULLANI (Ivix): Bio-Ecological |
| KNOW | 14 2002 | Diversity vs. Socio-Economic Diversity. A Comparison of Existing Measures Madda JANGGENS and Chair STEVAERT (Init). The pairs of Diversity within Operation Studies Deleter and |
| KNOW | 14.2003 | Maddy JANSSENS and Chris STEYAERT (Ivix): Theories of Diversity within Organisation Studies: Debates and |
| KNOW | 15.2003 | Future Trajectories Trajec PAYCAN LEVENT Fune MASUREL and Poton NIIV AMP (hin): Diversity in Entrepreneurable: Ethnic |
| KNOW | 13.2003 | Tuzin BAYCAN LEVENT, Enno MASUREL and Peter NIJKAMP (lvix): Diversity in Entrepreneurship: Ethnic and Female Roles in Urban Economic Life |
| KNOW | 16.2003 | Alexandra BITUSIKOVA (Ivix): Post-Communist City on its Way from Grey to Colourful: The Case Study from |
| KNOW | 10.2003 | Slovakia |
| KNOW | 17.2003 | Billy E. VAUGHN and Katarina MLEKOV (Ivix): A Stage Model of Developing an Inclusive Community |
| KNOW | 18.2003 | Selma van LONDEN and Arie de RUIJTER (Ivix): Managing Diversity in a Glocalizing World |
| Coalition | 19.2003 | Sergio CURRARINI: On the Stability of Hierarchies in Games with Externalities |
| Theory | 17.2003 | Sergio Containanti. On the Stating of Pretarenes in Gaines with Externatives |
| Network | | |
| PRIV | 20.2003 | Giacomo CALZOLARI and Alessandro PAVAN (lvx): Monopoly with Resale |
| PRIV | 21.2003 | Claudio MEZZETTI (lvx): Auction Design with Interdependent Valuations: The Generalized Revelation |
| | | Principle, Efficiency, Full Surplus Extraction and Information Acquisition |
| PRIV | 22.2003 | Marco LiCalzi and Alessandro PAVAN (lvx): Tilting the Supply Schedule to Enhance Competition in Uniform- |
| | | Price Auctions |
| PRIV | 23.2003 | David ETTINGER (lvx): Bidding among Friends and Enemies |
| PRIV | 24.2003 | Hannu VARTIAINEN (lvx): Auction Design without Commitment |
| PRIV | 25.2003 | Matti KELOHARJU, Kjell G. NYBORG and Kristian RYDQVIST (lvx): Strategic Behavior and Underpricing in |
| | | <u>Uniform Price Auctions: Evidence from Finnish Treasury Auctions</u> |
| PRIV | 26.2003 | Christine A. PARLOUR and Uday RAJAN (lvx): Rationing in IPOs |
| PRIV | 27.2003 | Kjell G. NYBORG and Ilya A. STREBULAEV (lvx): Multiple Unit Auctions and Short Squeezes |
| PRIV | 28.2003 | Anders LUNANDER and Jan-Eric NILSSON (lvx): Taking the Lab to the Field: Experimental Tests of |
| | | Alternative Mechanisms to Procure Multiple Contracts |
| PRIV | 29.2003 | TangaMcDANIEL and Karsten NEUHOFF (lvx): <u>Use of Long-term Auctions for Network Investment</u> |
| PRIV | 30.2003 | Emiel MAASLAND and Sander ONDERSTAL (lvx): Auctions with Financial Externalities |
| ETA | 31.2003 | Michael FINUS and Bianca RUNDSHAGEN: A Non-cooperative Foundation of Core-Stability in Positive |
| | | Externality NTU-Coalition Games |
| KNOW | 32.2003 | Michele MORETTO: Competition and Irreversible Investments under Uncertainty |
| PRIV | 33.2003 | Philippe QUIRION: Relative Quotas: Correct Answer to Uncertainty or Case of Regulatory Capture? |
| KNOW | 34.2003 | Giuseppe MEDA, Claudio PIGA and Donald SIEGEL: On the Relationship between R&D and Productivity: A |
| PT. | 25 2002 | Treatment Effect Analysis Alexander DEL BOCA Manife CALEOTTI and Banks BOTA Non-convenition in the Adjustment of Different |
| ETA | 35.2003 | Alessandra DEL BOCA, Marzio GALEOTTI and Paola ROTA: Non-convexities in the Adjustment of Different |
| GG | 36.2003 | Capital Inputs: A Firm-level Investigation Matthieu GLACHANT: Voluntary Agreements under Endogenous Legislative Threats |
| PRIV | 36.2003 | Narjess BOUBAKRI, Jean-Claude COSSET and Omrane GUEDHAMI: Postprivatization Corporate |
| 1 1/1 / | 31.2003 | Governance: the Role of Ownership Structure and Investor Protection |
| | | |

- (l) This paper was presented at the Workshop "Growth, Environmental Policies and Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, June 1, 2001
- (li) This paper was presented at the Fourth Toulouse Conference on Environment and Resource Economics on "Property Rights, Institutions and Management of Environmental and Natural Resources", organised by Fondazione Eni Enrico Mattei, IDEI and INRA and sponsored by MATE, Toulouse, May 3-4, 2001
- (lii) This paper was presented at the International Conference on "Economic Valuation of Environmental Goods", organised by Fondazione Eni Enrico Mattei in cooperation with CORILA, Venice. May 11, 2001
- (liii) This paper was circulated at the International Conference on "Climate Policy Do We Need a New Approach?", jointly organised by Fondazione Eni Enrico Mattei, Stanford University and Venice International University, Isola di San Servolo, Venice, September 6-8, 2001
- (liv) This paper was presented at the Seventh Meeting of the Coalition Theory Network organised by the Fondazione Eni Enrico Mattei and the CORE, Université Catholique de Louvain, Venice, Italy, January 11-12, 2002
- (lv) This paper was presented at the First Workshop of the Concerted Action on Tradable Emission Permits (CATEP) organised by the Fondazione Eni Enrico Mattei, Venice, Italy, December 3-4, 2001 (lvi) This paper was presented at the ESF EURESCO Conference on Environmental Policy in a Global Economy "The International Dimension of Environmental Policy", organised with the collaboration of the Fondazione Eni Enrico Mattei, Acquafredda di Maratea, October 6-11, 2001
- (lvii) This paper was presented at the First Workshop of "CFEWE Carbon Flows between Eastern and Western Europe", organised by the Fondazione Eni Enrico Mattei and Zentrum fur Europaische Integrationsforschung (ZEI), Milan, July 5-6, 2001
- (lviii) This paper was presented at the Workshop on "Game Practice and the Environment", jointly organised by Università del Piemonte Orientale and Fondazione Eni Enrico Mattei, Alessandria, April 12-13, 2002
- (Ivix) This paper was presented at the ENGIME Workshop on "Mapping Diversity", Leuven, May 16-17, 2002
- (lvx) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by the Fondazione Eni Enrico Mattei, Milan, September 26-28, 2002

2002 SERIES

CLIM Climate Change Modelling and Policy (Editor: Marzio Galeotti)

VOL *Voluntary and International Agreements* (Editor: Carlo Carraro)

SUST Sustainability Indicators and Environmental Valuation

(Editor: Carlo Carraro)

NRM Natural Resources Management (Editor: Carlo Giupponi)

KNOW *Knowledge, Technology, Human Capital* (Editor: Dino Pinelli)

MGMT *Corporate Sustainable Management* (Editor: Andrea Marsanich)

PRIV Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)

ETA *Economic Theory and Applications* (Editor: Carlo Carraro)

2003 SERIES

CLIM Climate Change Modelling and Policy (Editor: Marzio Galeotti)

GG Global Governance (Editor: Carlo Carraro)

SIEV Sustainability Indicators and Environmental Valuation

(Editor: Anna Alberini)

NRM Natural Resources Management (Editor: Carlo Giupponi)

KNOW Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)

IEM *International Energy Markets* (Editor: Anil Markandya)

CSRM Corporate Social Responsibility and Management (Editor: Sabina Ratti)

PRIV Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)

ETA Economic Theory and Applications (Editor: Carlo Carraro)

CTN *Coalition Theory Network*